\documentclass{article}
\usepackage{amsmath,amsfonts}
%\usepackage[OT4,OT1,T1]{fontenc}
%\usepackage[latin1,cp1255]{inputenc}
\usepackage{array}
\makeatletter
%\usepackage{babel}
%\newcommand{\theauthor}[1]{\hfill\begin{otherlanguage}{english}{\it #1}\end{otherlanguage} }
%\font\scf=ptmr8r scaled 910
%\font\kukua=ptmr8r scaled 1370
%\font\kukub=ptmr8r scaled 1200
%\usepackage{ij}
%\usepackage{times}
\newcommand{\rF}{\operatorname{F}}
\newcommand{\rG}{\operatorname{G}}
\newcommand{\rSp}{\operatorname{Sp}}
\newcommand{\bbR}{\mathbb{R}}

\textheight=550pt
\def\ {\thinspace}
\lineskiplimit=100pt 
\lineskip=9pt

\def\index\indexvolume#1\indexyear#2\endindextitle{
\rm\parindent=0pt%\overfullrule=0pt
\centerline{ISRAEL JOURNAL OF MATHEMATICS}
\vskip0.4cm
\centerline{TABLE OF CONTENTS, VOLUME #1\unskip, #2\unskip}
\vskip1.2cm
\frenchspacing}
\def\endindex{}


%\def\indexitem\indexauthor#1\indextitle#2\indexpage#3\endindexitem
%{\halign {\normalbaselines%
%\vbox{\hsize=11cm\hangindent=2em\hangafter=1 ##}%
%&\hbox to 1.3cm{\hfill ##}\cr%
%{\sc #1\unskip}\quad{\sl #2} & #3 \cr}\par}
%\def\indexauthor{}
%\def\endindexitem{}
%\def\indexpage{}
%\def\indextitle{}
\newcommand{\indexitem}[3]{\halign {\normalbaselines%
\vbox{\hsize=11cm\hangindent=2em\hangafter=1 ##}%
&\hbox to 1.3cm{\hfill ##}\cr%
{\sc #1\unskip}\quad{\sl #2} & #3 \cr}\par}

%\def\indexsee\indexauthor#1\partner#2\endindexitem
%{{\sc #1\unskip\enspace}{\sl See }{\sc #2\unskip}\par}

\newcommand{\indexsee}[2]{{\sc #1\unskip\enspace}{\sl See }{\sc #2\unskip}\par}

\makeatother

\begin{document}
\pagestyle{empty}

\index
\indexvolume 162
\indexyear 2007
\endindextitle

%3454 5
\indexitem{Ab\'{e}rt, M. and P.\ Heged\H us}{On the number of generators needed for free profinite products of finite groups}{81}

%3424 16
\indexitem{Bre\v sar, M.}{Commutativity preserving maps revisited}{317}

%3451 11
\indexitem{Deng, S. and Z.\ Hou}{On symmetric Finsler spaces}{197}

%3439 2
\indexitem{Deville, R. and P.\ H\'ajek}{Smooth noncompact operators from $C(K)$, $K$ scattered}{29}

%3506 4
\indexitem{Fern\'andez-Alcober, G.\ A.}{Omega subgroups of powerful $p$-groups}{75}

%3389 6
\indexitem{De Filippis, V.}{An Engel condition with generalized derivations on multilinear polynamials}{93}

%3439 2
\indexsee{H\'ajek P.}{Deville, R.}

%3531 19
\indexitem{Harbater, D.}{Correction and addendum to ``Embedding problems with local conditions''}{373}

%3454 5
\indexsee{Heged\H us, P.}{M.\ Ab\'{e}rt}

%3419 11
\indexsee{Hou, Z.}{S.\ Deng}

%3431 10
\indexitem{Irmak, E. and M.\ Korkmaz}{Automorphisms of the Hatcher-Thurston complex}{183}

%3429 9
\indexitem{Kalenda, O.\ F.\ K.}{(I)-envelopes of closed convex sets in Banach spaces}{157}

%3443 15
\indexitem{Kalton, N.\ J.}{Extending Lipschitz maps into $\mathcal{C}(K)$-spaces}{275}

%3433 13
\indexitem{Koll\'ar, J.}{A conjecture of Ax and degenerations of Fano varieties}{235}

%3431 10
\indexsee{Korkmaz, M.}{E.\ Irmak}

%3440 17
\indexitem{Li, B.\ Qin and E.\ G.\ Saleeby}{On solutions of functional-differential equations $f'(x)=a(x)f(g(x))+b(x)f(x)+c(x)$ in the large}{335}

%3436 7
\indexitem{Li, Y.-C., R.\ Sato and S.-Y.\ Shaw}{Convergence theorems and Tauberian theorems for functions and sequences in Banach spaces and Banach lattices}{109}

%3435 14
\indexitem{Natale, S.}{On the exponent of tensor categories coming from finite groups}{253}

%3423 18
\indexitem{Rosendal, C. and S.\ Solecki}{Automatic continuity of homomorphisms and fixed points on metric compacta}{349}

%3440 17
\indexsee{Saleeby, E.\ G.}{B.\ Qin Li}

%3436 7
\indexsee{Sato, R.}{Y.-C.\ Li}

%3436 7
\indexsee{Shaw, S.-Y.}{Y.-C.\ Li}

%3423 18
\indexsee{Solecki, S.}{C.\ Rosendal}

%3442 3
\indexitem{Spinas, O.}{Splitting squares}{57}


%3489 8
\indexitem{de~Th\'elin, H.}{Construction of measures with expansion}{151}

%3432 12
\indexitem{Wang, X.}{Algebraic convergence theorems of $n$-dimensional Kleinian groups}{221}

%3428 1
\indexitem{Yekutieli, A.}{Mixed resolutions and simplicial sections}{1}







\end{document}